

# JGR Solid Earth

# **RESEARCH ARTICLE**

10.1029/2022JB023953

# **Key Points:**

- Paleopoles do not average "out" paleosecular variation, which forms a first-order contributor to the dispersion of coeval paleopoles
- The number of sites used to compute a paleopole and the number of paleopoles calculated from a given data set is essentially arbitrary
- Calculating apparent polar wander paths from site-level data instead of poles allows the weighting of uncertainties and the amount of data

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

B. Vaes, b.vaes@uu.nl

### Citation:

Vaes, B., Gallo, L. C., & van Hinsbergen, D. J. J. (2022). On pole position: Causes of dispersion of the paleomagnetic poles behind apparent polar wander paths. *Journal of Geophysical Research: Solid Earth*, *127*, e2022JB023953. https://doi. org/10.1029/2022JB023953

Received 4 JAN 2022 Accepted 18 MAR 2022

© 2022. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

### and continent

1. Introduction

and continents relative to the Earth's rotation axis. They provide the reference frame for paleogeography, paleoclimate, and paleoenvironment studies, and serve as the reference against which we compare paleomagnetic data collections obtained from deformed terranes to assess relative latitudinal motions or vertical-axis rotations (e.g., Besse & Courtillot, 2002; Butler, 1992; Creer et al., 1954; Irving, 1964; Torsvik et al., 2008, 2012; van Hinsbergen et al., 2015). APWPs are commonly constructed as a sequence of reference poles of different age that are calculated from a collection of coeval paleomagnetic poles (hereafter "paleopoles"). These paleopoles are assumed to represent stable plate interiors and to provide an accurate representation of the time-averaged geomagnetic field, which is in turn assumed to approximate a geocentric axial dipole (GAD; e.g., Butler, 1992; Creer et al., 1954; Tauxe & Kent, 2004). In the absence of errors in the data, the input paleopoles are expected to

Apparent polar wander paths (APWPs) are computed to constrain the past position and motion of tectonic plates

# On Pole Position: Causes of Dispersion of the Paleomagnetic Poles Behind Apparent Polar Wander Paths

Bram Vaes<sup>1</sup>, Leandro C. Gallo<sup>2,3</sup>, and Douwe J. J. van Hinsbergen<sup>1</sup>

<sup>1</sup>Department of Earth Sciences, Utrecht University, Utrecht, The Netherlands, <sup>2</sup>Instituto de Geociencias Básicas, Aplicadas y Ambientales de Buenos Aires (IGEBA), Buenos Aires, Argentina, <sup>3</sup>Centre for Earth Evolution and Dynamics (CEED), University of Oslo, Oslo, Norway

**Abstract** Paleomagnetic poles used to compute apparent polar wander paths (APWPs) are strongly dispersed, which was recently shown to cause a large fraction (>50%) of these poles to be statistically distinct from the APWP to which they contributed, suggesting that current statistical approaches overestimate paleomagnetic resolution. Here, we analyze why coeval paleopoles are so dispersed, using the paleopoles behind the most recent global APWP and a compilation of paleomagnetic data obtained from <10 Ma volcanic rocks (PSV10). We find that paleopoles derived from sedimentary rocks, or from data sets underrepresenting paleosecular variation (PSV), are more dispersed and more frequently displaced. We show that paleopoles based on a smaller number of paleomagnetic sites are more dispersed than poles based on larger data sets, revealing that the degree to which PSV is averaged is an important contributor to the pole dispersion. We identify as a fundamental problem, however, that the number of sites used to calculate a paleopole, and thus the dispersion of coeval paleopoles, is essentially arbitrary. We therefore explore a different approach in which reference poles of APWPs are calculated from site-level data instead of paleopoles, thereby assigning larger weight to larger data sets. We introduce a bootstrap-based method for comparing a collection of paleomagnetic data with a reference data set on the same hierarchical level, whereby the uncertainty is weighted against the number of paleomagnetic sites. Finally, our study highlights that demonstrating smaller tectonic displacements requires larger paleomagnetic data sets, and that such data sets can strongly improve future APWPs.

**Plain Language Summary** Apparent polar wander paths (APWPs) are widely used to reconstruct the position of continents relative to the Earth's rotation axis. These paths are typically calculated by averaging paleomagnetic poles obtained from rocks of similar age. Although these poles are expected to be tightly grouped, they are strongly scattered. Notably, this causes >50% of the poles used in recent APWPs to be statistically different from the APWP itself. Here, we investigate to what extent errors in these poles may explain the observed scatter. We find that poles derived from sedimentary rocks are more scattered than those derived from igneous rocks. Also, poles based on smaller data sets are more dispersed than those based on larger data sets. Our analysis shows that the amount of paleomagnetic data used to determine a pole is often arbitrary. To overcome the subjectivity in pole calculation, we propose a new approach in which an APWP is calculated from individual data points instead of from paleopoles. This allows comparing paleomagnetic data on the same statistical level, and the development of APWPs in which larger data sets have larger weight. Our study thus emphasizes the value of collecting large paleomagnetic data sets, which may improve future APWPs.





**Figure 1.** Orthographic projection of the 50–70 Ma paleopoles (white circles, with 95% confidence ellipses ( $A_{95}$ ) in gray) used to compute the 60 Ma reference pole (black diamond) of the global APWP of Torsvik et al. (2012). The  $A_{95}$  and  $K_{95}$  associated with the reference pole are 2.1° and 14.0° and are indicated by the filled and unfilled black circles. The paleopole derived from the Vaigat Formation (Riisager et al., 2003; red diamond, with  $A_{95} = 9.2°$  in red) is shown as an example of a statistically "displaced" paleopole. The VGPs used to calculate this paleopole are plotted as red squares. Please note that all poles and VGPs are shown in south pole coordinates in a South African coordinate frame.

plot close to, and be statistically indistinguishable from, the reference pole of the APWP to which they contribute (Butler, 1992; Rowley, 2019; Tauxe et al., 2010). But they do not.

Harrison and Lindh (1982) and, more recently, Rowley (2019) demonstrated that with commonly used comparison metrics, as much as 50% of the paleopoles used to compute APWPs are statistically distinct from the reference pole to which they contributed (Figure 1). Rowley (2019) argued that it is inappropriate to determine tectonic displacements from the angular difference and combined uncertainty of a paleopole and a reference pole, as is typically done, because these poles and their confidence limits are calculated on a different hierarchical level (e.g., Bazhenov et al., 2016). Instead, Rowley (2019) pointed out that a geologically meaningful difference can only be interpreted from individual paleopoles that lie beyond the angular dispersion of the paleopoles that underlie the APWP. He proposed that the uncertainty of an APWP should then be determined by the angular distance from the APWP that includes 95% of the input paleopoles (his  $K_{95}$ , see Equation 2 in Rowley, 2019), which for the widely used global APWP of Torsvik et al. (2012) is often as large as ~15–20° (Figure 1). In this approach, the resolution of an APWP is governed by the dispersion of the input paleopoles. Rowley (2019) thus emphasized that the reliable use and future improvement of APWPs requires thorough understanding of the nature and causes of the dispersion of input poles and their possible displacement relative to the APWP.

The sources of uncertainty that likely contribute to the dispersion of paleopoles have so far mostly been sought in the inadequate estimation of individual pole positions (e.g., Domeier et al., 2012; Harrison & Lindh, 1982; Kent & Muttoni, 2020; Van der Voo, 1990), caused by for example, inadequate representation of paleosecular variation (PSV), unremoved magnetic overprints, age uncertainties, measurement errors, inclination shallowing in sediments, or unrecognized deformation (e.g., Butler, 1992; Harrison & Lindh, 1982; Rowley, 2019; Tauxe et al., 2010; Vaes et al., 2021). In addition, the dispersion may be enhanced by nondipole contributions to the paleomagnetic field, or by errors in the relative plate motion reconstructions that were used to transfer paleopoles to a common coordinate frame (e.g., Butler, 1992; Domeier et al., 2012; Tauxe et al., 2010). But even though the problem of the large dispersion of paleopoles has long been acknowledged (e.g., Bazhenov et al., 2016; Harrison & Lindh, 1982; Van der Voo, 1990), how the above sources of scatter, and any other, contribute to the dispersion of paleopoles has not been systematically analyzed.



# Journal of Geophysical Research: Solid Earth



**Figure 2.** (a) Orthographic projection of the global APWP of Torsvik et al. (2012) in South African coordinates (orange diamonds, each shown with their  $A_{95}$ ), computed using a running mean approach with a time window of 20 Ma. The input paleopoles are plotted as circles that are colored by age. (b) Histogram of the number of input paleopoles per 10 Ma time interval (after Torsvik et al., 2012).

In this study, we investigate to what extent uncertainties and errors in the paleopoles themselves, as well as in the way these poles are defined, may explain pole dispersion. We further examine why a large fraction of those poles is statistically displaced from the APWP to which they contribute using classical comparison metrics. To this end, we analyzed high-quality data sets of volcanic rocks from the last 10 Ma (PSV10; Cromwell et al., 2018), in which plate reconstruction uncertainties, tectonic deformation-induced deviations, and common pitfalls related to sedimentary rocks such as inclination shallowing, play no significant role. We first use this data set to evaluate how the dispersion of paleopoles is influenced by the extent to which PSV is sampled. Then, we analyze which other factors may further enhance the dispersion, and which factors may cause so many paleopoles to be displaced from the reference pole to which they contributed. For this purpose, we also used the compilation of paleopoles behind the global APWP of Torsvik et al. (2012) (Figure 2). Based on our analyses, we explore an optimal approach to calculate APWPs and their use to quantify tectonic displacements, in such a way that statistical differences with an APWP can be used to draw geologically meaningful conclusions.

# 2. Background: Current Ways to Calculate and Use APWPs

The calculation and use of APWPs rely on the fundamental assumption that the time-averaged geomagnetic field approximates that of a GAD field, whereby the time-averaged geomagnetic pole coincides with the Earth's rotation axis. Through this assumption, APWPs describe the apparent wandering of the rotation axis relative to a geographical reference location through geological time and provide a paleomagnetic reference frame for the paleogeographic reconstruction of continents and geological terranes (e.g., Butler, 1992). APWPs are typically constructed as a sequence of paleomagnetic poles that each represent the geomagnetic pole position relative to a reference (usually a continent, e.g., Africa) for a chosen geological time interval (Figure 2a).

The first APWP was constructed by Creer et al. (1954), who connected a time series of individual paleopoles obtained from rocks of the British Isles. When more paleopoles became available, reference poles were calculated for selected geological periods by averaging all paleopoles obtained from rocks of that period (e.g., Irving, 1964; Phillips & Forsyth, 1972; Van der Voo & French, 1974). In the last four decades, most APWPs have been constructed using paleopoles as input, in one of two ways: either by a running mean method (e.g., Irving, 1977; Van Alstine & de Boer, 1978), or by fitting a smoothed spline path through a selection of paleopoles (e.g., Jupp & Kent, 1987; Thompson & Clark, 1981, 1982). The presently most widely used reference APWPs were calculated



using the running mean method (e.g., Besse & Courtillot, 2002; Torsvik et al., 2012). The main advantage of this method is that it is relatively simple and intuitive, allowing the APWP to be straightforwardly reproduced by other workers. In the running mean approach, a series of reference poles is calculated at a chosen time interval (e.g., 10 Ma) by averaging collections of paleopoles whose mean rock age falls within a chosen time window (e.g., 20 Ma; e.g., Besse & Courtillot, 2002; Kent & Irving, 2010; Torsvik et al., 2012, 2008; Wu et al., 2021). The input paleopoles may either be derived from a single continent or tectonic plate, yielding a continental/ single plate APWP, or from poles obtained from multiple continents and plates after rotating them to a common coordinate frame using relative plate reconstructions, yielding a global (or "master") APWP (e.g., Besse & Courtillot, 1991, 2002; Phillips & Forsyth, 1972; Torsvik et al., 2008, 2012).

The paleopoles used to calculate an APWP typically represent a "study mean" paleomagnetic pole, corresponding to the main result of an individual paleomagnetic study (e.g., Hospers, 1954). On the other hand, there are also studies that produce multiple paleopoles and sometimes poles obtained in different studies of the same sedimentary or volcanic sequence are averaged to a single paleopole (e.g., Torsvik et al., 2012). A single paleopole is typically obtained through averaging a set of virtual geomagnetic poles (VGPs) that are each calculated from a site-mean paleomagnetic direction through the assumption of a GAD field (Figure 1; see Chapter 7 in Butler (1992) for a detailed explanation). A paleomagnetic "site" is defined as a geological unit that represents an increment of geological time relative to the timescale over which the Earth's magnetic field changes (following McElhinny & McFadden, 2000), such that the corresponding VGP represents a "spot reading" of the field. Whether or not a rock unit represents an increment of geological time is up to the interpretation of the paleomagnetist: lava flow units are typically assumed to provide such spot readings (e.g., Biggin et al., 2008; Cromwell et al., 2018; Johnson et al., 2008). But whether paleomagnetic directions from sediments provide spot readings, either at sample level or at sedimentary "bed" level, is more subjective (e.g., Meert et al., 2020; Tauxe & Kent, 2004; Vaes et al., 2021). Paleomagnetists aim to acquire a collection of independent spot readings of the paleomagnetic field that is sufficiently large and contains enough geological time to "average out" PSV. But the minimum number of VGPs (N) required to accurately represent PSV is unclear and depends on the time interval between different sites and the behavior of the geomagnetic field at the time. In practice, the number of VGPs used to compute a paleopole is mostly arbitrary and is often determined by the availability of suitable rocks for sampling, the sampling strategy, or the number of sites or samples that were interpreted to provide reliable paleomagnetic directions. Gerritsen et al. (2022) recently showed the benefits of collecting as many paleomagnetic sites as possible for the determination of a robust paleopole, and emphasized that sampling efforts are better spent on collecting more paleomagnetic sites than on collecting more samples per site. In his widely used list of reliability criteria, Van der Voo (1990) advised using at least 25 paleomagnetic samples to calculate a paleopole, whereas the recently updated "Van der Voo" criteria of Meert et al. (2020) suggested using  $N \ge 8$  paleomagnetic sites (each representing a spot reading) for the calculation of a paleopole. On the other hand, a statistical analysis of Tauxe et al. (2003) showed that N > 100 may be required for a robust representation of PSV. In practice, each individual paleopole that is included in the calculation of an APWP is assumed to provide an accurate estimate, within its confidence limits, of the time-averaged geomagnetic pole relative to the sampling location, regardless of N.

If the paleopoles obtained from rocks of similar age indeed provide accurate estimates of the time-averaged field, then the scatter of paleopoles around the mean pole should only be caused by nondipole contributions to the geomagnetic field and plate reconstruction errors (e.g., Butler, 1992; Irving, 1964; Rowley, 2019). In the running mean approach, the pole dispersion may also result, in part, from the apparent polar wander that occurred during the time window (e.g., 20 Ma) from which the input poles have been included (e.g., Van der Voo, 1990). The position of paleopoles may differ substantially if their ages differ by, for example, >10 Ma, especially when those poles are derived from rapid-moving tectonic plates. By incorporating multiple paleopoles in the calculation of a reference pole of an APWP, paleomagnetists aim to average "out" this scatter (e.g., Irving, 1964), irrespective of its causes. The uncertainty in the position of the reference pole is traditionally expressed by the 95% cone of confidence about the pole ( $A_{95,ref}$ ), which is computed using Fisher (1953) statistics under the implicit assumption that the input paleopoles conform to a Fisher (1953) distribution (Heslop & Roberts, 2019; Rowley, 2019). In this approach, it is common to assign unit weight to each input paleopole, regardless of the uncertainty in the position or age of the pole, or the number of VGPs used to calculate it (e.g., Besse & Courtillot, 2002; Irving & Irving, 1982; Morel & Irving, 1981; Torsvik et al., 2008, 2012).



But already 40 years ago, Harrison and Lindh (1982) noted that the scatter of poles obtained from stable plate interiors is much larger than expected (see Figures 1 and 2a) and consequently argued against the uniform weighting of paleopoles in the calculation of APWPs. They suspected that the paleopole scatter resulted, for a large part, from the incomplete averaging of PSV in many paleopoles. To overcome this problem, they proposed a method for calculating APWPs in which individual paleopoles are weighted according to for example, the number of underlying VGPs and the age range of the sampled rocks. Over the years, different weighting schemes have been proposed, in which individual paleopoles are weighted either quantitatively or against a set of qualitative criteria (e.g., Gallo et al., 2021; Hansma & Tohver, 2020; Harrison & Lindh, 1982; Le Goff et al., 1992; McFadden & McElhinny, 1995; Musgrave, 1989; Schettino & Scotese, 2005; Swanson-Hysell et al., 2019; Thompson & Clark, 1981; Torsvik et al., 1996, 2008, 2012, 1992; Wu et al., 2021). Some authors calculated the reference poles of single-continent APWPs by averaging VGPs rather than paleopoles (Hansma & Tohver, 2020; McElhinny et al., 1974; McElhinny & McFadden, 2000; van Hinsbergen et al., 2017), thereby assigning larger weight to larger data sets of independent measurements of the past geomagnetic field. Because the weighting of poles based on a chosen set of criteria inevitably introduces more subjectivity (e.g., Irving & Irving, 1982; Morel & Irving, 1981), the currently most widely used APWPs are still based on a running mean approach in which all poles are weighted equally and in which errors are not propagated (Besse & Courtillot, 2002; Torsvik et al., 2012).

One of the prime applications of APWPs is the identification and quantification of tectonic displacements of tectonic blocks or plates relative to a certain reference continent (see Chapter 11 of Butler (1992) for an extensive review). To determine a tectonic displacement—expressed either as a relative vertical-axis rotation or latitudinal displacement—paleomagnetists typically compare an individual paleopole derived from the studied tectonic block with a reference APWP. If the paleopole is shown to be statistically distinct from the APWP, according to some comparison metric, this is typically interpreted as evidence for a relative vertical-axis rotation and/or latitudinal displacement. To assess whether a paleopole is statistically distinct (or "discordant") with respect to the reference pole (Butler, 1992; Rowley, 2019), paleomagnetists typically compare the 95% confidence regions about the paleopole with that of a reference pole, even though these are of a different statistical "rank" (following the terminology of Bazhenov et al., 2016). The uncertainty in the position of the individual paleopole ( $A_{95,PP}$ ) is typically calculated from the distribution of VGPs using Fisher (1953) statistics (see Heslop and Roberts (2019) for a review). In the classical approach, a paleopole is considered statistically displaced if its angular distance to the reference pole exceeds the combined uncertainty of the paleopole and the reference pole, calculated using:

$$\sqrt{A_{95,ref}^2 + A_{95,PP}^2} \tag{1}$$

whereby the  $A_{95,ref}$  and the  $A_{95,PP}$  are the radius of the 95% confidence circle around the reference pole and the independent paleopole, respectively. It is important to note that the uncertainty in the position of the underlying paleopoles (their  $A_{95,PP}$ ) is typically not propagated in the calculation of the  $A_{95,ref}$ , which may thus be an underestimate of the uncertainty in the reference pole position (Heslop & Roberts, 2020).

Rowley (2019) showed that, with the above equation, >50% of the paleopoles used in the global APWP of Torsvik et al. (2012) are statistically displaced relative to the APWP itself. He pointed out that a significant difference between an independent paleopole and a reference pole can therefore not be straightforwardly interpreted as a signal of tectonic motion. Instead, he argued, significant tectonic motion can only be concluded if the independent paleopole falls outside the circle that contains 95% of the input paleopoles from which the mean poles of the APWP is computed (which he referred to as the  $K_{95}$ ). This confidence circle was previously used by Bazhenov et al. (2016) to define "alarm bands" around APWPs, which they used to identify potential remagnetization events. Using the  $K_{95}$  as a confidence estimate of the APWP, rather than the  $A_{95,ref}$  essentially means that the resolution at which we can discern relative tectonic displacements is for a large part determined by the angular deviation of the paleopoles that lie furthest from the mean pole of the APWP. Increasing the resolution of APWPs for tectonic purposes would then require decreasing the scatter of input poles around the APWP. Below, we therefore analyze what causes the scatter of poles behind an APWP, and what causes their displacement relative to the mean pole to which they contributed.



# 3. Data and Methods

We conduct a series of experiments to explore and identify the main contributors to the dispersion of paleopoles that are used as input for current APWPs and to determine why a large fraction of poles are statistically distinct in the traditional statistical framework (see Equation 1)—from the APWP to which they contribute. With this aim, we assess the influence on the pole dispersion, and pole displacement, of a range of parameters associated with each input pole, such as the number of independent measurements of the geomagnetic field used to calculate the pole (N), lithology, the uncertainties in the pole position itself and the age uncertainty of the sampled rocks.

We use two data sets: the PSV10 database, and the compilation of paleopoles used for the Torsvik et al. (2012) global APWP. The PSV10 database contains 83 paleomagnetic data sets (from 81 publications) obtained from volcanic rocks that formed in stable plate interiors in the last 10 Ma, as compiled by Cromwell et al. (2018). This data set is the most recent update of a compilation of paleomagnetic data that fulfill all commonly used quality criteria of volcanic rocks that erupted so recently that age uncertainty, plate motion, tectonic deformation, or artifacts common in sedimentary rocks such as inclination shallowing, are considered negligible. We note that Cromwell et al. (2018) corrected all site locations and VGPs for recent plate motions using the no-net-rotation model of Argus et al. (2011), which only involved small adjustments of up to a few degrees. The PSV10 compilation and its predecessors (e.g., PSVRL; McElhinny & McFadden, 1997) are typically used to study the behavior of the recent geomagnetic field, including paleosecular variation, and serves as the standard of statistical models of the geomagnetic field such TK03.GAD (Tauxe & Kent, 2004) and BCE19 (Brandt et al., 2020). Each VGP included in the compilation has been interpreted as a nearly "instantaneous" reading of the magnetic field and is calculated by averaging multiple paleomagnetic directions obtained from the same lava flow unit to decrease the effect of (small) measurement errors. We use the "original" PSV10 database (i.e., not corrected for serial correlation, see Section 4 of Cromwell et al., 2018) to assess the influence of representation of PSV on paleopole scatter around a mean for the geomagnetic field of the last 10 Ma. We calculated a paleopole from each paleomagnetic data set included in PSV10 using Fisher (1953) statistics, providing a pole position and associated  $A_{05}$  and Fisher precision parameter K (Table S1).

We quantify the dispersion of VGPs or paleopoles using three different statistical parameters: the Fisher (1953) precision parameter K, the angular standard deviation (or VGP scatter) S, and the mean angular distance to the reference S'. S is a widely used measure of the scatter of VGPs and is commonly used to assess PSV in studies of geomagnetic field behavior (see Equation 14.1, Tauxe et al., 2010). S' provides an alternative measure of the dispersion and has the advantage that it is less affected by outliers (Doubrovine et al., 2019; Suttie et al., 2015). In addition, we tested whether the paleopoles whose data were used in the PSV10 database would be deemed statistically displaced according to the traditional comparison metric of Equation 1. Because tectonic motions of lithology-based artifacts are considered irrelevant for the PSV10 data set, we use the observed scatter of poles about the "true" mean pole to evaluate the effect of incomplete averaging of PSV, or nondipole contributions to the geomagnetic field.

Next, we explore to what extent other common sources of error in paleomagnetic data may have further enhanced the dispersion of paleopoles that were used to calculate the global APWP of Torsvik et al. (2012). We first used to entire database of 501 paleopoles that were obtained from 0 to 330 Ma old rocks to assess the magnitude of pole dispersion, and whether this varies with age and between different lithological groups. Then, for a selection of 167 paleopoles that fall within the 0–110 Ma age range, we compiled additional characteristics of each paleomagnetic data set including the number of paleomagnetic sites used to calculate the pole, the uncertainties in the pole position, and in the age uncertainty of the sampled rocks (Table S2). To assess the influence of specific parameters on the dispersion of individual paleopoles, we used the angular distance of each paleopole to the mean pole of the APWP, following for example, Harrison and Lindh (1982). It is worth noting that the mean poles of the global APWP do not represent the "true" time-averaged pole position, but they provide a useful approximation of the time-averaged pole and are therefore used as the reference poles in our analyses. In addition, to quantify the clustering of poles that fall within a specific time interval of, for example, 20 Ma, we used the Fisher (1953) precision parameter, analogous to K for distributions of VGPs. We performed these calculations using a set of inhouse-developed Python codes, for which we made extensive use of the functions and programs from the freely available paleomagnetic software package PmagPy (Tauxe et al., 2016). The Python codes used for this study are publicly available through the online repository on Zenodo (Vaes, 2022; see Data Availability Statement).





**Figure 3.** (a) Orthographic projection of the VGPs included in the PSV10 database (Cromwell et al., 2018). The paleopoles computed from the 83 data sets of PSV10 are plotted as red circles. The reference poles calculated from the VGPs and paleopoles are shown by the white square and orange star, respectively. (b) Histogram of the angular distance to their reference pole of the VGPs and paleopoles shown in (a). (c) Angular distance to the mean pole of the 83 paleopoles versus the number of VGPs (*N*). The mean angular distance and 95% deviation of the 5,000 synthetic paleopoles, which are computed for every second value of *N*, are depicted by black squares (see text for explanation). A log-linear regression curve computed for the paleopoles included in PSV10 is shown in blue (with bootstrapped 95% confidence regions). (d) Plot of the angular distance of each paleopole against the combined uncertainty, as calculated using Equation 1 (after Rowley, 2019). Poles that do not satisfy the Deenen et al. (2011) criterion or whereby the underlying distribution of VGPs is not consistent with a Fisher (1953) distribution are plotted as red circles or blue triangles, respectively.

### 4. Results

# 4.1. Analysis of PSV10

The 83 paleomagnetic data sets included in the PSV10 database (Table S1) vary in size from N = 3 to N = 128 VGPs, with an average N of ~29, and from each of these collections of VGPs we computed a paleopole (Figure 3a). We calculated the time-averaged "reference" geomagnetic pole for the last 10 Ma by computing the Fisher (1953) mean of all 2,401 VGPs, as well as by averaging the paleopoles without taking their uncertainties into account, as is common when computing APWPs (Figure 3a). The angular distance between the two references poles (0.6°) is statistically insignificant; the VGP-based mean has an  $A_{95,VGPs} = 0.7^{\circ}$  and pole-based mean has  $A_{95,poles} = 1.2^{\circ}$ . We

use these data-based reference poles to enable the application of the comparison metrics and to ensure consistency with the analyses in the next section. We note that using the geographic pole instead of these reference poles hardly influences the results described below (see Figure S1 in Supporting Information S1).

As a measure of the dispersion of VGPs and paleopoles, we calculated the angular distance of each VGP and each paleopole to the respective reference pole positions (Figure 3b). The dispersion of the paleopoles, each averaging a different number of VGPs that were obtained within a single study, is, logically, smaller than that of the individual VGPs. The total distribution of VGPs has K = 17.1,  $S = 20.1^{\circ}$ , and  $S' = 15.6^{\circ}$ , compared to the distribution of paleopoles which has K = 172.3,  $S = 6.2^{\circ}$ , and  $S' = 5.2^{\circ}$ . The 95% angular deviation of the population of VGPs is 33.8° (calculated using Equation 2 in Rowley, 2019), whereas that of the population of paleopoles, corresponding to the  $K_{95}$ , is 10.7°. So even though averaging collections of VGPs to paleopoles considerably decreases the scatter, some of the "study mean" paleopoles are still significantly dispersed around the reference pole (Figures 3a and 3b).

To assess to what extent PSV underlies this pole scatter, we first conducted a numerical simulation in which we generated sets of synthetic paleomagnetic poles by randomly sampling (with replacement) N VGPs from the PSV10 database. For every second value of N (from 2 to 70) we generated 5,000 pseudosamples from which we calculated synthetic poles along with their angular distance to the reference pole. We note that the magnitude of VGP dispersion is latitude-dependent, at least for the last 10 Ma (e.g., Biggin et al., 2008; Cox, 1962, 1970; Doubrovine et al., 2019; McFadden et al., 1991), but because paleopoles used to calculate APWPs are often derived from a wide latitude range, we draw VGPs from the entire data set in our experiment. Figure 3c shows how the scatter of these synthetic paleopoles decreases with *N*, with the mean angular distance to the reference pole (*S'*) gradually decreasing from almost 9° for N = 4 to ~2° for N = 70. The observed angular distances of the paleopoles calculated from the PSV10 data sets clearly follow the trend obtained for the synthetic poles (Figure 3c). Also, we find that the majority (75%) of the paleopoles that are at a relatively large angular distance to the reference pole, that is larger than the mean value of  $5.2^{\circ}$  (*S'*), have N < 25 (Figure 3c). Overall, these results illustrate that PSV is never averaged "out": larger data sets simply provide a more accurate estimate of the grand mean pole position.

Next, we estimated the expected pole dispersion resulting from PSV by randomly distributing the 2,401 VGPs over 83 collections of VGPs with the same *N* as the data sets that contributed to PSV10. We then determined the mean angular distance to the reference pole (*S'*) of the 83 paleopoles that were calculated from those 83 collections of VGPs. We repeated this procedure 5,000 times and obtained an average *S'* of 3.9° (with 95% bootstrap confidence bounds of 3.4° and 4.4°) instead of the 5.2° obtained from the published data sets. This suggests that the majority (~4°) of the dispersion of poles in the PSV10 database is a signal of PSV and that a relatively small contribution to the scatter (~1.3°) results from other artifacts that may affect a data set of a particular study (e.g., local deformation, erroneous structural corrections, unremoved magnetic overprints, measurement errors).

Finally, we tested whether the poles calculated for the PSV10 data set are statistically displaced from the mean pole position, using Equation 1. We find that 26.5% (22 out of 83) of the paleopoles that are used in the PSV10 data set are, under this comparison statistic, distinct from the mean pole (Figure 3d). This result cannot be entirely attributed to the relatively low  $A_{95}$  of 1.2°; even if we would use an  $A_{95}$  of 3.0° instead, which is a typical  $A_{95}$  value for a mean pole of the global APWP of Torsvik et al. (2012), then 15.7% (13 out of 83) of the poles would still be displaced. We find that the data set used to compute three of the 22 displaced poles may have undersampled PSV, as indicated by their  $A_{95}$  values that are smaller than the lower limit of the  $A_{95min-max}$  envelope of Deenen et al. (2011) (Figure 3d and Table S1). In addition, seven of the displaced poles do not conform to a Fisher (1953) distribution, as indicated by the quantile-quantile method of Fisher et al. (1987). The other 12 poles pass all common reliability criteria, however. If we would exclude all data sets that do not pass these additional criteria, we find that 23.5% is statistically displaced according to Equation 1. This illustrates that even in the PSV10 data set, which should contain little artifacts, a significant fraction of poles is statistically displaced from the reference pole calculated from all 83 paleopoles.



# Journal of Geophysical Research: Solid Earth



**Figure 4.** (a) Angular distance to the reference pole of all 0–330 Ma paleopoles used to compute the global APWP (in South African coordinates) of Torsvik et al. (2012). The reference poles are computed as running mean poles with a 20 Ma time window. The mean angular distance and  $K_{95}$  computed from the paleopoles included in each 20 Ma time window are plotted as black diamonds and blue squares, respectively. The mean angular distance is plotted in red if the distribution of paleopoles is not consistent with a Fisher (1953) distribution. (b) Histogram of the angular distance to the reference pole for all paleopoles in the database. *Note.* Because most paleopoles are used twice in the running mean approach, the number of poles in this figure is higher than the actual number of paleopoles (501).

### 4.2. Analysis of the Paleopoles Used for the Global APWP

The mean (median) angular distance to the reference pole of the 962 paleopoles (most poles count twice) used for the global APWP for the last 320 Ma of Torsvik et al. (2012) is 8.0° ( $6.9^{\circ}$ ; Figure 4 and Table S2). This is higher than for the PSV10 data set, suggesting that there are additional sources of scatter. This trend is also obvious from the 95% angular deviation ( $K_{95}$ ) of the total population of paleopoles of 18.2° (Figure 4a). The magnitude of pole dispersion is relatively constant with time with an average angular distance of ~7°, and up to 9–10° between 140–200 and 280–330 Ma (Figure 4b). The  $K_{95}$  of the paleopoles for the majority of 20 Ma time windows is around 12–15°, but the time windows with increased dispersion yield  $K_{95}$  values of up to ~20° (Figure 4b). For most time windows, the application of the quantile-quantile method of Fisher et al. (1987) indicates that the input paleopoles are consistent with a Fisher (1953) distribution around the reference pole (Figure 4b). This suggests that the contribution of polar wander to the pole dispersion, which is expected to cause an elliptical distribution of the paleopoles within a time window, is relatively small. We also find that using a smaller time window of 10 Ma instead of 20 Ma only leads to a minor decrease in the pole dispersion (mean and median angular distance of 7.5° and 6.5°, see Figure S2 in Supporting Information S1), which supports the above interpretation. In the following, we assess the potential relationships between the observed dispersion of paleopoles and specific characteristics of the paleopoles and specific characteristics of the paleopoles.

The adequate sampling and averaging of the PSV are key requirements for determining an accurate paleopole (Deenen et al., 2011; Meert et al., 2020; Van der Voo, 1990). Paleomagnetic poles based on a larger number of independent measurements of the geomagnetic field should provide a more accurate estimate of the time-averaged geomagnetic field and thus lie closer to the position of the "true" time-averaged geomagnetic pole. We compiled the number paleomagnetic sites (*N*) used for the calculation of the 167 paleopoles included for the 0–110 Ma segment of the global APWP of Torsvik et al. (2012). The number of sites used to calculate a pole varies widely between different studies, from N = 3 to N = 507, with a median value of 20 (Figure 5a). For sedimentary rocks, the number of sites may represent individual samples, but sometimes also averages of multiple samples from one or multiple beds. Notably, 12% of the poles (20 in total) are calculated from less than the minimum of eight paleomagnetic sites formulated by Meert et al. (2020) in their recent update of the Van der Voo (1990) criteria. Figure 5b shows the angular distance of each paleopole against the number of sites used to compute that pole. For this compilation of paleopoles, we observe a general decrease of the angular distance of paleopoles to the reference pole with increasing *N*, from ~8° for N < 8 to ~5° for N > 50. This trend is similar to that observed for the PSV10 database (Figure 3c) and reflects the degree to which PSV is represented by the





**Figure 5.** (a) Histogram of the number of VGPs (N) used to compute each of the 0–110 Ma paleopoles behind the global APWP of Torsvik et al. (2012). (b) Angular distance of each paleopole to the reference pole versus N. Each paleopole is plotted by their lithology. The reference poles are computed as running mean poles with a 20 Ma time window. The median angular distance is computed for 15 ranges of N and shown as purple diamonds. A log-linear regression curve (blue, with bootstrapped 95% confidence region) highlights the trend.

data. The mean angular distance is, however, a few degrees higher which may reflect additional sources of scatter, which we analyze below.

Contrary to the PSV10 database, the global APWP also contains poles from intrusive igneous rocks, carbonate rocks, and clastic sedimentary rocks. Differences exist between these broad lithological groups in terms of how they record the past geomagnetic field and the properties and robustness of paleomagnetic data derived from them. Clastic sedimentary rocks are typically considered to be less reliable paleomagnetic recorders than igneous rocks because of their weaker remanent magnetization and the common shallowing of the paleomagnetic inclination, caused by syndepositional and postdepositional processes (e.g., Bilardello et al., 2013; King, 1955; Kodama, 2012; Tauxe & Kent, 2004; Vaes et al., 2021). The mean (median) angular distance of paleopoles derived from clastic sedimentary rocks is indeed larger  $(9.2^{\circ} (8.0^{\circ}))$  than those derived from igneous rocks  $(7.0^{\circ} (6.5^{\circ}); Figure 6)$ .

Torsvik et al. (2012) corrected the poles derived from clastic sedimentary rocks by applying a so-called "blanket" flattening factor of f = 0.6, except for poles that were calculated from data sets on which the original authors already applied an inclination shallowing correction. We find that the application of this "blanket" correction factor does not significantly reduce the dispersion of paleopoles derived from clastic sedimentary rocks. The paleopoles that were corrected with this flattening factor have a mean (median) angular distance of 9.5° (8.2°), whereas the uncorrected poles have a mean (median) angular distance of 9.7° (8.3°) to the reference pole of the APWP, which was recalculated using the uncorrected sediment-derived paleopoles. We note that Torsvik et al. (2012) did not apply an inclination shallowing correction to paleopoles derived from carbonate rocks, even though they may also be affected by inclination error (e.g., Muttoni et al., 2013; Vaes et al., 2021). The number of paleopoles in the Torsvik et al. (2012) database that were derived either from carbonate rocks or from clastic sediments that were corrected for inclination shallowing by the original authors is, however, too low for a meaningful calculation of the mean angular distance.

Next, we evaluate the relationship between the dispersion of paleopoles and the commonly used statistical parameters *K* and  $A_{95}$ . The range of *K* values thought be representative for PSV is ~10–50 (Deenen et al., 2011) or 10–70 (Meert et al., 2020) and collections of VGPs with higher values likely underrepresent PSV. The angular distance of igneous-based poles relative to the reference pole is ~6° at *K* values of 20–40, increasing to a slightly higher value of ~7° at *K* values higher than 80 (Figure 7a). We also find that the angular distance increases with increasing  $A_{95}$  (Figure 7b). The  $A_{95}$  confidence circle is a function of both *K* and *N* and considering the observed



# Journal of Geophysical Research: Solid Earth



**Figure 6.** Histograms of the angular distance to the reference pole of all 0–330 Ma paleopoles that are derived from igneous and carbonate rocks (a) and clastic sedimentary rocks (b). The poles derived from the latter rocks are corrected for inclination shallowing using a "blanket" flattening factor of f = 0.6, following Torsvik et al. (2012). The mean (median) values are indicated by the vertical straight (dashed) lines.

trends in *K* (Figure 7a) and *N* (Figure 5b), these results suggests that although the dispersion increases with a tighter clustering of VGPs, this effect is counteracted by the decrease in dispersion with increasing *N*. These combined effects may explain the relatively low mean angular distance for paleopoles with relatively small  $A_{95}$  values.

There does not seem to be a clear correlation between the angular distance and age uncertainty of paleopoles (Figure 7c), whether derived from igneous or sedimentary rocks. We note, however, that the rocks sampled to compute the 0-110 Ma paleopoles have age uncertainties rarely exceeding 15 Ma, which may be higher for poles from lower Mesozoic and Paleozoic rocks. We also find no strong relationship between the difference in age of the reference pole and the mean age of the input paleopole (Figure S2 in Supporting Information S1), indicating that the effect on the pole dispersion of polar wander on timescales of <10 Ma is relatively small.

A common selection criterion to include poles into an APWP is the *Q*-factor of Van der Voo (1990), whose underlying quality criteria were recently updated by Meert et al. (2020). The *Q*-factor ranges from 1 to 7, indicating how many of each of the seven quality criteria of Van der Voo (1990) are satisfied by the paleomagnetic data set. In the calculation of some recent APWPs, such as the spline paths of Torsvik et al. (2012) and the APWPs of Wu et al. (2021), larger statistical weight was assigned to paleopoles with a higher *Q*-factor compared to those with low *Q*. We used the entire data set of Torsvik et al. (2012) from 0 to 330 Ma, which only included paleopoles with  $Q \ge 3$ , to assess whether there is a correlation between the *Q*-factor and its angular distance to the mean pole to which it contributes. We find no correlation between the quality factor *Q* and the angular distance of these poles (Figure 7d). Our finding confirms the analysis by Van der Voo (1990) based on his compilation of Phanerozoic poles from Europe and North America, based on which he showed that there was no substantial decrease in the mean angular distance for Q > 2 (see Figure 7 of Van der Voo, 1990).

According to the traditional comparison approach (Equation 1), we find that 52.2% of the 0–110 Ma paleopoles used for the 0–100 Ma segment of the global APWP of Torsvik et al. (2012) is statistically different from the reference pole to which they contribute, that is, the angular distance to the reference pole exceeds the combined uncertainty of both poles (Figure 8). Paleopoles derived from clastic sedimentary rocks are not only more scattered around the reference pole to which they contribute (Figure 6) but are also more often statistically displaced than paleopoles derived from igneous rocks (Figure 8a). Nearly 80% (30 of 39) of the clastic sedimentary poles are displaced compared to ~48% (113 of 235) of the igneous poles.





**Figure 7.** Angular distance to the reference pole for all 0–110 Ma paleopoles in the compilation of Torsvik et al. (2012) against the Fisher (1953) precision parameter K (a),  $A_{95}$  (b) and the age uncertainty range (c). The median angular distance is computed for several intervals and shown as purple diamonds. A log-linear regression curve (blue, with bootstrapped 95% confidence region) highlights the trend for each parameter. (d) Boxplots showing the median angular distance and its distribution of the 0–330 Ma paleopoles behind the global APWP with a given *Q*-factor (Van der Voo, 1990).

Finally, of all paleopoles that have an  $A_{95}$  that falls outside of the *N*-dependent reliability envelope of Deenen et al. (2011), which assesses whether the scatter is straightforwardly explained by PSV alone, 72% is found to be statistically displaced (Figure 8b), mostly because of low scatter of the underlying VGPs, yielding an  $A_{95} < A_{95,min}$ . This supports the conclusion of Harrison and Lindh (1982) that underrepresentation of PSV by the paleomagnetic data set is an important cause of individual paleopoles to be statistically displaced from the reference pole to which they contribute. Together with the trend that poles with higher *K* values have larger angular distances to the reference pole, this shows that undersampling of PSV increases the dispersion of paleopoles used in current APWPs, and the number of statistically displaced poles.

### 5. Discussion

### 5.1. The Problem of Using Paleopoles

Rowley (2019) argued that the traditional method used to determine a statistically significant difference between independent paleopoles and an APWP is flawed, because a large fraction of input poles is statistically "displaced" from the reference pole to which they contribute. Such statistical differences therefore do not necessarily represent the tectonically meaningful difference that is commonly interpreted from them (e.g., Butler, 1992; Coe et al., 1985; Harrison & Lindh, 1982; Rowley, 2019). What causes these paleopoles to be displaced (according to the traditional statistical framework)? Our analysis in part confirms the widely held suspicion that common errors





Figure 8. Plots of the angular distance of each paleopole against the combined uncertainty, as calculated using Equation 1 (after Rowley, 2019). Poles are colored by lithology (a) or by whether the underlying VGPs conform to a Fisher (1953) distribution (b).

and uncertainties in paleomagnetic data contribute to this. We find that paleopoles are more often statistically distinct when derived from data sets in which PSV is likely underrepresented (indicated by high *K* values (e.g., K > 70; Meert et al., 2020), and an  $A_{95}$  i.e., below the  $A_{95,min}$  of the reliability envelope of Deenen et al. (2011)), or from clastic sedimentary rocks, regardless of whether they are corrected for inclination shallowing using a "blanket" correction factor of 0.6 or not (*sensu* Torsvik et al., 2012). Excluding paleopoles based on such data sets may thus decrease the dispersion of paleopoles. We also find, however, that the dispersion of paleopoles, and their statistical difference with the reference pole to which they contribute, cannot solely be attributed to the "insufficient" quality of the paleopoles themselves.

Our analysis reveals that even when paleomagnetic data sets pass stringent and widely applied quality criteria, the paleopoles derived from them are frequently statistically different from the reference pole to which they contribute. Our analysis of 0-110 Ma paleopoles behind the APWP of Torsvik et al. (2012) shows that ~50% of the igneous-based paleopoles, that have a high Q-factor and that satisfy the criterion of Deenen et al. (2011), are statistically displaced from the reference pole to which they contribute. Moreover,  $\sim 26\%$  of the paleopoles obtained from the high-quality data sets of recently erupted lavas included in PSV10 are statistically displaced from the mean pole calculated for the recent (<10 Ma) geomagnetic field. These findings illustrate that the accuracy at which we can determine the time-averaged geomagnetic pole position using a single paleopole is not as high as our statistical metrics suggest, and that the formal confidence region (the  $A_{os}$ ) may underestimate the true uncertainty of an individual paleopole (e.g., Coe et al., 1985; Harrison & Lindh, 1982; Rowley, 2019). This may be partly explained by the fact that uncertainties are typically not propagated through each hierarchical level in standard paleomagnetic analyses (Heslop & Roberts, 2020). Our results show that there are often sources of scatter that cause poles to be displaced that cannot be, or at least were not, recognized by the paleomagnetists that collected the data sets, and that cannot be straightforwardly identified using commonly applied quality criteria. Such sources of dispersion may include nondipole contributions to past geomagnetic field, errors in tilt correction and/or unrecognized intraplate deformation (e.g., Besse & Courtillot, 2002; Butler, 1992). Overall, our results support the notion from Rowley (2019) that the observed dispersion may be inherent to paleomagnetic data, and the paleopoles derived from them. But because the dispersion of coeval paleopoles is not accounted for in the traditional statistical framework, Rowley (2019) proposed an alternative approach in which the uncertainty of an APWP is directly determined by the dispersion of input poles, irrespective of the sources of the dispersion. The magnitude of dispersion depends, however, on subjective choices, which we illustrate below.

Our results show that the number of VGPs used to compute paleopoles (N) is a first-order contributor to the dispersion of coeval paleopoles (Figures 3c and 5a). And this introduces a fundamental problem: there is no definition of the amount of paleomagnetic data that defines paleopole. Widely used quality criteria only include a minimum amount of data needed to calculate a reliable paleopole: Van der Voo (1990) proposed a minimum of 25 samples, to which Meert et al. (2020) recently added that these samples should preferably be derived from  $N \ge 8$ paleomagnetic sites. Because these values only provide a minimum, this allows the calculation, and inclusion in an APWP, of paleopoles based on a highly variable number of paleomagnetic data, as illustrated by the poles used in the most recent global APWP (Figure 5a). Although the number of VGPs behind a paleopole is often simply determined by the amount of "reliable" paleomagnetic data presented in a single study, multiple paleopoles from the same study are sometimes included as separate poles in the calculation of an APWP, or a single "grand" mean pole is computed from data obtained in different studies of the same volcanic or sedimentary sequence. For example, in the global APWP of Torsvik et al. (2012), a Cretaceous paleomagnetic pole from Mongolia was calculated from N = 126 lava sites (van Hinsbergen et al., 2008), whereas 42 sites from North American intrusive rocks of approximately the same age (McEnroe, 1996) were used to compute five different input poles of N = 5-12(Figure 9a). Likewise, that APWP includes seven poles from seven published data sets of the Deccan Traps of India constrained by N varying from 3 to 130 (Figure 9b), whereas these data sets could instead be combined into a single mean pole for the Deccan Traps, as done by Vandamme et al. (1991).

If the number of paleomagnetic sites/VGPs (N) behind a single paleopole is essentially arbitrary, as illustrated above, and N is a first-order contributor to the dispersion of poles, then the  $K_{95}$  value is dependent on arbitrary choices as well. Given a data set of VGPs, we may calculate reference poles of an APWP with low  $A_{95}$  and high  $K_{95}$  by including many paleopoles based on low N, or with high  $A_{95}$  and low  $K_{95}$  by combining the same data over a few paleopoles with high N. Below we will therefore explore the alternative avenue of calculating a reference APWP from VGP-level data directly, instead of from paleopoles that each represent an average of an arbitrarily defined collection of VGPs.

### 5.2. Toward a VGP-Based Comparison Method

Calculating a reference data set from VGPs instead of paleopoles allows the comparison with an independent paleomagnetic data set of VGPs on the same hierarchical level and avoids subjective choices regarding the amount of paleomagnetic data that is used to determine a single paleopole. VGP-based APWPs have been constructed before, for example, Australia (Hansma & Tohver, 2020; McElhinny et al., 1974), Iberia (van Hinsbergen et al., 2017), and Laurentia (Swanson-Hysell et al., 2019), and for all major continents by McElhinny and McFadden (2000). Also, paleomagnetic data from different studies and locations have been combined on the VGP level in studies of geomagnetic field behavior (e.g., Brandt et al., 2021; Cromwell et al., 2018). Calculating APWPs from VGPs poses several challenges, however. First, a method needs to be designed to avoid the "displacement" problem highlighted by Rowley (2019). Second, for many published paleomagnetic data sets, individual VGPs have not been reported and estimates need to be developed from published statistical descriptions of the data.

The problem of displaced paleopoles that follows from the application of the traditional comparison metric (Equation 1) will be even more prominent for VGP-based APWPs. Because the total number of VGPs (*N*) is much higher than the number of paleopoles, the  $A_{95}$  calculated from all individual VGPs is even smaller than that calculated from the associated paleopoles. With so many paleopoles already statistically displaced from the APWP using classical comparison metrics (see Equation 1), this problem will also exist or be even larger for VGP-based APWPs. To illustrate this: for the PSV10 data set, the  $A_{95}$  calculated from the "study mean" poles that contributed to the PSV10 data set is  $1.2^{\circ}$ , and from the VGPs is  $0.7^{\circ}$ . Using the VGP-based  $A_{95}$  as the  $A_{95,ref}$  instead of the pole-based  $A_{95}$ , the number of displaced poles determined with Equation 1 increases from 22 to 23. This shows that a different measure needs to be developed that ensures that ~95% of the input data sets do not statistically differ from the reference data set. Below, we use the PSV10 data set to explore such an alternative approach.

Our analysis shows that the angular difference between a paleopole and a reference pole decreases with an increasing number of VGPs (N; Figures 3c and 5a). We therefore introduce a new comparison metric between





**Figure 9.** Examples of how subjective choices related to the definition and inclusion of certain paleopoles may influence the reference pole position and associated parameters. See text for discussion. (a) Orthographic projection of the 100-120 Ma paleopoles used to compute the 110 Ma reference pole of the global APWP of Torsvik et al. (2012). Two reference poles are plotted: computed from all paleopoles (black star) or if the McEnroe (1996) poles are replaced by a "study mean" pole (green star). The difference between the reference poles is  $1.2^{\circ}$ . (b) Same as (a), but this time showing the 50-70 Ma paleopoles used to compute the 60 Ma reference pole. Again, two reference poles are plotted: computed from all paleopoles (black star) or if the Deccan Traps poles are replaced by the Deccan Superpole of Vandamme et al. (1991). The difference between the reference poles is  $1.0^{\circ}$ .

paleopoles and reference data sets that takes *N* into account. Central to this approach is an alternative expression for the confidence limit of the reference pole (of an APWP), which we refer to as the " $B_{95}$ ." The  $B_{95}$  is an estimate of the 95% confidence limit that the reference data set, represented by a reference pole, would have had if it had been constrained by the same N as the studied paleopole. In other words, the  $B_{95}$  represents a prediction of the pole position of a paleopole with given N, if that paleopole would have sampled N VGPs from the total cloud of "reference" VGPs. This way, the resolution of a comparison between a reference data set and a collection of VGPs





**Figure 10.** Example of the proposed VGP-based comparison method (see text for discussion). (a) Orthographic projection of the paleomagnetic data set of Oliva-Urcia et al. (2016) that was included in PSV10. The VGPs and the paleopole computed from those VGPs are plotted as red circles and the red star (with  $A_{95}$ ), respectively. To assess whether this paleopole is statistically distinct from the reference data set, 2,000 pseudopoles are computed by randomly drawing the same number of VGPs (N = 17) from the total data set (small black dots). The  $B_{95}$  is then defined as the 95% angular deviation of these pseudopoles from the principal vector of the distribution of pseudopoles and is indicated by the black circle. (b) Cumulative distribution of the pseudopoles plotted against the angular distance from the reference pole. Vertical lines show the angular distance of the paleopole, the  $B_{95}$  and the combined uncertainty computed (i) following the traditional framework (see Equation 1) and (ii) by replacing the  $A_{95,ref}$  with the  $B_{95}$ . This figure indicates that the studied paleopole is not statistically "displaced" according to our new metric, whereas it is displaced according to the traditional statistical framework. (c) Bootstrap common mean test of Tauxe et al. (2010) applied to the studied data set (blue), showing that the data set is not statistically different on the 95% confidence level from the set of pseudopoles (red). The three subplots each show the cumulative distribution of bootstrapped means in the three Cartesian coordinates x, y, and z. (d) Plot of the angular distance of each paleopole against the combined uncertainty for all data sets included in PSV10, whereby the combined uncertainty is computed after replacing the  $A_{95,ref}$  in Equation 1 by the  $B_{95}$  (see also Table S1). Using the VGP-based approach, the number of displaced poles decreases from 22 to 7 (out of 83).

obtained by a paleomagnetic study is directly dependent on the *N* of the latter data set, and thus naturally reflects the amount of information included in the statistical comparison.

To determine the  $B_{95}$ , we use a bootstrap approach: for each bootstrap run we compute a pseudopole from N randomly drawn VGPs from the reference data cloud. By repeating this procedure a few thousand times, we obtain thousands of pseudopoles, each derived from the same amount of VGPs as included in the studied pale-omagnetic data set. Next, we calculate the  $B_{95}$  as the angular distance from the principal vector of the cloud of pseudopoles that contains 95% of the pseudopoles (Figures 10a and 10b). Evidently, the  $B_{95}$  becomes larger when these pseudopoles are calculated from a smaller number of VGPs and is thus directly proportional to the number of VGPs included in the studied data set. This approach makes no a priori assumptions about the behavior or statistical properties of the geomagnetic field, but simply predicts where poles based on N VGPs could lie, as a function of the scatter in the reference data set, regardless of the sources of this scatter.



We now assess how this equal-*N* approach performs by determining the number of "displaced" paleomagnetic data sets included in the PSV10 database. To this end, we compute the  $B_{95}$  for each "study mean" pole using its underlying number of VGPs, and then replace the  $A_{95,ref}$  by the  $B_{95}$  in Equation 1. Again, we consider a data set statistically distinct from the reference if the angular distance of the study mean pole exceeds the combined uncertainty, which in this approach is a function of the  $B_{95}$  of the reference and the  $A_{95}$  of the compared paleopole. We find that this comparison method reduces the percentage of displaced poles from 26.5% to 8.4% (7 out of the 83 data sets, Figure 10d). Some of these data sets are thus still "displaced," also according to our proposed comparison metric. We note, however, that four of these data sets were almost entirely derived from a sequence of successive lava flows, which led Cromwell et al. (2018) to discard most of the VGPs in their filtered PSV10 data sets, since those VGPs may not have provided entirely independent spot readings of the field.

The above results suggest that replacing the  $A_{95,ref}$  with the bootstrapped  $B_{95}$  in Equation 1 may provide an alternative solution for the "displacement" problem pointed out by Rowley (2019). The application of this approach requires, however, that the reference data set (of an APWP) is calculated from VGPs, which are not always available.

### 5.3. Parametric Resampling of VGPs

Paleomagnetic data sets available for computation of APWPs have not always been reported on the VGP level, but only per paleopole, and the vast majority of published paleomagnetic data has not been included in modern paleomagnetic databases (e.g., MagIC, Jarboe et al., 2012; Paleomagnetism.org, Koymans et al., 2020). To overcome this problem, we may parametrically resample VGPs from the paleopoles given the published statistical properties of these poles. Similar approaches of parametrically resampling paleopoles or VGPs have been frequently used for paleomagnetic data analyses and simulations (e.g., Cromwell et al., 2018; Gallo et al., 2021; Koymans et al., 2016; Rowley, 2019; Smirnov & Tarduno, 2010; Swanson-Hysell et al., 2014; Tauxe et al., 1991). To evaluate whether this approach is appropriate here, we reproduce the PSV10 data set by parametrically resampling VGPs from the "study mean" poles and their statistical parameters (referred to as a "parametric bootstrap," following Tauxe et al., 2010). For each data set included in PSV10, we created the same number of VGPs as included in the published data set from a Fisher (1953) distribution around the "study mean" pole and its associated precision parameter K, which were both calculated from the published VGPs. The parametrically sampled data cloud is very similar to that of the published VGPs (Figure 11a), with little difference in the associated statistical parameters. We observe a slightly larger dispersion of the parametrically sampled VGPs (indicated by a lower K and higher S' value) compared to the published VGPs (Figure 11b). This is explained by the outlying VGPs in the published data sets that lower the K value for that data set, such that when these data sets are parametrically resampled, a larger fraction of the simulated VGPs are between  $\sim 15^{\circ}$  and  $\sim 50^{\circ}$  from the mean pole of all VGPs (Figure 11a).

We now test whether the data sets that contributed to the PSV10 database are displaced relative to a reference data set consisting of parametrically resampled VGPs instead of the published VGPs. From the collection of 2401 parametrically sampled VGPs, we compute a pseudopole by drawing the required *N*, which is again equal to the *N* that is behind the investigated data set included in PSV10. We then repeated this procedure 2,000 times, each time generating a new set of parametrically sampled VGPs, to obtain the set of pseudopoles from which we compute the  $B_{95}$ . We find that this  $B_{95}$  is typically ~0.1–0.4° larger than that obtained from the published VGPs (Table S1). This illustrates that using parametrically sampled VGPs instead of the "real" VGPs is slightly more conservative, making it thus slightly more difficult to demonstrate a significant displacement between a paleopole and the reference data set. We find that using parametrically sampled VGPs to compute the  $B_{95}$  makes no difference for the data sets included in PSV10: the number of "displaced" paleopoles remains the same (7 out of 83; Table S1). This therefore opens the opportunity to compute future APWPs, including a global APWP, from VGP-level data, even if all the original VGPs are not available.

### 5.4. Outlook: Future APWPs and Their Application

We show here that calculating APWPs from VGP-level data may provide an alternative solution to the "displacement" problem in current pole-based APWPs, whereby subjective choices related to the definition of individual





**Figure 11.** Comparison between the VGPs included in the PSV10 compilation and the parametrically sampled VGPs derived from the paleopoles and their statistical parameters. (a) Orthographic projection of the VGPs included in PSV10 (blue dots) and parametrically sampled VGPs (red dots) from one simulation whereby these VGPs were generated for each data set of PSV10 from a Fisher (1953) distribution around the paleopole and described by the *K* value of the underlying VGPs. (b) Histogram of the angular distance from the reference pole of the published VGPs (blue) and parametrically sampled VGPs (obtained from 100 simulations). The mean angular distance of the simulated VGPs (red line) is slightly larger than that of the published VGPs (blue line).

paleopoles are avoided and paleomagnetic data can be compared with a chosen reference on the same hierarchical level. Computing APWPs from VGPs has more benefits: it allows the incorporation and propagation of key uncertainties and errors in paleomagnetic data, such as age uncertainty, as demonstrated by the bootstrap or Monte Carlo-based methods recently developed by Swanson-Hysell et al. (2019) and Hansma and Tohver (2020). The uncertainty in the position of a "study mean" paleopole, which is determined by the number and scatter of the underlying VGPs, is automatically accounted for in a VGP-based approach. To incorporate age uncertainty, a reference pole and its  $B_{95}$  can be computed from only those VGPs that fall in a chosen age range. Age uncertainty could then be weighed by parametrically sampling VGPs from all reference paleopoles that overlap in age with the chosen time window and assigning a random age within the age range of the reference paleopole to each VGP. Each pseudopole from which the reference pole position and its  $B_{95}$  is computed is then drawn from only those VGPs whose ages fall within the time window. When comparing an independent paleomagnetic data set with a VGP-based APWP, the age range may be equated to the age range of the sampled rocks, thereby accounting for both the number of VGPs that data set and its age uncertainty.

On the other hand, the VGP-based approach also comes with a necessity for larger data scrutiny, particularly for large data sets. In the VGP-approach unit weight is assigned to each VGP, such that largest weight is assigned to largest paleomagnetic data sets. This essentially means that the accuracy of each paleopole is assumed to be a function of N alone; with increasing N, PSV and other sources of noise are increasingly averaged. However, some paleomagnetic data sets may be affected by systematic bias that is unrelated to N, such as local deformation, errors in tilt correction, systematic magnetic overprints, a local/regional magnetic field anomaly, inclination shallowing, and undersampling of PSV. The majority of input paleopoles of the global APWP of Torvsik et al. (2012) that are displaced according to classical comparison metrics likely suffer from one or more of these artifacts. To identify a potential bias in future VGP-based APWPs caused by such artifacts requires a careful analysis of statistical differences between different input data sets, with particular focus on the data sets with largest *N*. In general, it will be important to apply stringent and objective selection criteria, such as those from Meert et al. (2020) and Vaes et al. (2021), for inclusion of data into the APWP.

We foresee that VGP-based APWPs, which have confidence intervals that better reflect the scatter of the underlying paleomagnetic data and do not depend on subjective choices, will provide more accurate and robust paleomagnetic reference frames, at shorter time intervals, to underpin paleogeography studies (e.g., van Hinsbergen et al., 2015) or true polar wander calculations (e.g., Kulakov et al., 2021; Muttoni & Kent, 2019; Torsvik



et al., 2012). At the same time, the  $B_{95}$  makes that the resolution of the reference is adjusted to the resolution of the paleomagnetic data set that is compared with the reference, such that demonstrating smaller statistical differences requires larger data sets. It also opens novel opportunities to compute high-resolution reference poles for specific times of rapid Earth system change by collecting large, well-dated paleomagnetic data sets. We emphasize that for VGP-based APWPs, it is of key importance that the paleomagnetic community makes all their published data publicly available, for example, in databases such as MagIC (Jarboe et al., 2012) or paleomagnetism.org (Koymans et al., 2020), as this will strongly improve future APWPs and contribute to solving detailed Earth system problems.

# 6. Conclusions

In this study, we investigated the causes of the large dispersion of paleopoles obtained from similar aged rocks that causes a large fraction of these poles to be statistically displaced from the mean pole to which they contribute. To this end, we used a previously compiled global data set from <10 Ma old volcanic rocks (PSV10) without tectonics-induced pole dispersion, and the compilation of paleopoles used to compute the most recent global APWP of Torsvik et al. (2012). Based on our analyses, we conclude the following:

- The dispersion of paleopoles, defined by their angular distance to the reference pole to which they contribute, decreases with an increasing number of VGPs from which they are calculated. Individual paleopoles, which are commonly based on a few dozen VGPs, do not average "out" PSV and their angular distance to the reference pole is, in part, dependent on the extent to which PSV has been sampled by the underlying paleomagnetic data set
- 2. Paleopoles that are statistically displaced from the reference pole of the APWP, according to the traditional statistical framework, are more often derived paleomagnetic data sets in which PSV is likely underrepresented, or from clastic sedimentary rocks, regardless of whether they are corrected for inclination shallowing using a "blanket" correction factor
- 3. The dispersion of paleopoles, and their statistical difference with a given reference pole, cannot only be attributed to the insufficient quality of the paleopoles. This is indicated by the absence of a correlation between the *Q*-factor of Van der Voo (1990) and the pole dispersion, as well as the high percentage (~26%) of displaced paleopoles derived from the data sets included in PSV10. The high percentage of displaced poles indicates that the formal confidence region (the  $A_{95}$ ) of individual paleopoles may underestimate the true uncertainty in pole position. This implies that sources of uncertainty cause poles to be displaced that are not straightforwardly identified by applying common quality criteria, which may include nondipole contributions, erroneous tilt corrections or unrecognized intraplate deformation
- 4. There is no clear definition of the amount of paleomagnetic data that underpins a single paleopole. Although all input paleopoles are typically assigned equal weight in APWP computation, they are an average of a collection of VGPs of essentially arbitrary size, and the number of VGPs that underlie paleopoles varies considerably. Also, the number of paleopoles derived from a single paleomagnetic data set or from a specific volcanic or sedimentary sequence is often determined by subjective choices. Such choices may potentially bias the position of reference poles of an APWP and the size of its confidence regions
- 5. Calculating a reference data set from VGP-level data avoids the problem of arbitrarily defined paleopoles. We explored a bootstrap-based approach to comparing a given paleomagnetic data set against such a VGP-based reference data set, whereby the resolution is determined by the number of VGPs in the studied data set. We show that this comparison metric significantly reduces the percentage of displaced data sets included in PSV10 (from ~24% to ~8%) and may thus provide an alternative solution to the problem of displaced paleopoles in current pole-based APWPs highlighted by Rowley (2019)
- 6. Constructing APWPs from VGP-level data instead of from paleopoles allows giving more weight to larger data sets and incorporating uncertainties on the position and in the age of the paleopole. This enables the comparison of paleomagnetic data sets with a reference APWP on the same hierarchical level, such that physically meaningful differences and associated uncertainties may be determined. Such an approach also implies that constraining smaller relative tectonic displacements requires larger, better dated paleomagnetic data sets. Moreover, future APWPs may thus be strongly improved by collecting large, high-quality data sets from stable plate interiors



# **Data Availability Statement**

No new paleomagnetic data are used in this study. The paleomagnetic data sets are derived from previous compilations by Torsvik et al. (2012) and Cromwell et al. (2018), and can be found both in the original publications as well as in the tables of the Supporting Information S1. The Python codes used for our analyses are publicly available on Zenodo (Vaes, 2022): https://doi.org/10.5281/zenodo.6366399.

# References

Argus, D. F., Gordon, R. G., & DeMets, C. (2011). Geologically current motion of 56 plates relative to the no-net-rotation reference frame. *Geochemistry, Geophysics, Geosystems*, 12, Q11001. https://doi.org/10.1029/2011GC003751

Bazhenov, M. L., Levashova, N. M., & Meert, J. G. (2016). How well do Precambrian paleomagnetic data agree with the Phanerozoic apparent polar wander path? A Baltica case study. *Precambrian Research*, 285, 80–90.

Besse, J., & Courtillot, V. (1991). Revised and synthetic apparent polar wander paths of the African, Eurasian, North American and Indian plates, and true polar wander since 200 Ma. Journal of Geophysical Research, 96(B3), 4029–4050.

Besse, J., & Courtillot, V. (2002). Apparent and true polar wander and the geometry of the geomagnetic field over the last 200 Myr. *Journal of Geophysical Research*, 107(B11), 2300. https://doi.org/10.1029/2000JB000050

Biggin, A. J., van Hinsbergen, D. J., Langereis, C. G., Straathof, G. B., & Deenen, M. H. (2008). Geomagnetic secular variation in the Cretaceous Normal Superchron and in the Jurassic. *Physics of the Earth and Planetary Interiors*, 169(1–4), 3–19.

Bilardello, D., Jezek, J., & Gilder, S. A. (2013). Role of spherical particles on magnetic field recording in sediments: Experimental and numerical results. *Physics of the Earth and Planetary Interiors*, 214, 1–13.

Brandt, D., Constable, C., & Ernesto, M. (2020). Giant Gaussian process models of geomagnetic palaeosecular variation: A directional outlook. *Geophysical Journal International*, 222(3), 1526–1541.

Brandt, D., Ernesto, M., & Constable, C. (2021). Consistent and contrasting aspects of the geomagnetic field across epochs with distinct reversal frequencies revealed by modeling the Kiaman Superchron. *Geochemistry, Geophysics, Geosystems*, 22, e2021GC009866. https://doi. org/10.1029/2021GC009866

Butler, R. F. (1992). Paleomagnetism: Magnetic domains to geologic terranes (Vol. 319). Boston, MA: Blackwell Scientific Publications.

Coe, R. S., Globerman, B. R., Plumley, P. W., & Thrupp, G. A., (1985). Paleomagnetic results from Alaska and their tectonic implications. In D. G. Howell (Ed.), *Tectonostratigraphic terranes of the Circum-Pacific region, Circum Pacific Earth Sciences Series* (Vol. 1, pp. 85–108). American Association of Petroleum Geologists.

Cox, A. (1962). Analysis of present geomagnetic field for comparison with paleomagnetic results. *Journal of Geomagnetism and Geoelectricity*, 13(3–4), 101–112.

Cox, A. (1970). Latitude dependence of the angular dispersion of the geomagnetic field. Geophysical Journal International, 20(3), 253-269.

Creer, K. M., Irving, E., & Runcorn, S. K. (1954). The direction of the geomagnetic field in remote epochs in Great Britain. Journal of Geomagnetism and Geoelectricity, 6(4), 163–168.

Cromwell, G., Johnson, C. L., Tauxe, L., Constable, C. G., & Jarboe, N. A. (2018). PSV10: A global data set for 0–10 Ma time-averaged field and paleosecular variation studies. *Geochemistry, Geophysics, Geosystems*, 19, 1533–1558. https://doi.org/10.1002/2017GC007318

Deenen, M. H., Langereis, C. G., van Hinsbergen, D. J., & Biggin, A. J. (2011). Geomagnetic secular variation and the statistics of palaeomagnetic directions. *Geophysical Journal International*, 186(2), 509–520.

Domeier, M., Van der Voo, R., & Torsvik, T. H. (2012). Paleomagnetism and Pangea: The road to reconciliation. Tectonophysics, 514, 14–43.

Doubrovine, P. V., Veikkolainen, T., Pesonen, L. J., Piispa, E., Ots, S., Smirnov, A. V., et al. (2019). Latitude dependence of geomagnetic paleosecular variation and its relation to the frequency of magnetic reversals: Observations from the Cretaceous and Jurassic. *Geochemistry*, *Geophysics, Geosystems*, 20, 1240–1279. https://doi.org/10.1029/2018GC007863

Fisher, N. I., Lewis, T., & Embleton, B. J. (1987). Statistical analysis of spherical data. Cambridge University Press.

Fisher, R. A. (1953). Dispersion on a sphere. Proceedings of the Royal Society of London-Series A: Mathematical and Physical Sciences, 217(1130), 295–305.

Gallo, L. C., Farjat, A. D., Tomezzoli, R. N., Calvagno, J. M., & Hernández, R. M. (2021). Sedimentary evolution of a Permo-Carboniferous succession in southern Bolivia: Responses to icehouse-greenhouse transition from a probabilistic assessment of paleolatitudes. *Journal of* South American Earth Sciences, 106, 102923. https://doi.org/10.1016/j.jsames.2020.102923

Gerritsen, D., Vaes, B., & van Hinsbergen, D. J. (2022). Influence of data filters on the position and precision of paleomagnetic poles: What is the optimal sampling strategy? *Geochemistry, Geophysics, Geosystems*, 23, e2021GC010269. https://doi.org/10.1029/2021GC010269

Hansma, J., & Tohver, E. (2020). Southward drift of eastern Australian hotspots in the paleomagnetic reference frame is consistent with global true polar wander estimates. *Frontiers of Earth Science*, 8, 544496. https://doi.org/10.3389/feart.2020.544496

Harrison, C. G. A., & Lindh, T. (1982). A polar wandering curve for North America during the Mesozoic and Cenozoic. Journal of Geophysical Research, 87(B3), 1903–1920.

Heslop, D., & Roberts, A. P. (2019). Quantifying the similarity of paleomagnetic poles. *Journal of Geophysical Research: Solid Earth*, 124, 12388–12403. https://doi.org/10.1029/2019JB018342

Heslop, D., & Roberts, A. P. (2020). Uncertainty propagation in hierarchical paleomagnetic reconstructions. Journal of Geophysical Research: Solid Earth, 125, e2020JB019488. https://doi.org/10.1029/2020JB019488

Hospers, J. (1954). Rock magnetism and polar wandering. Nature, 173(4416), 1183-1184.

Irving, E. (1964). Paleomagnetism and its application to geological and geophysical problems. New York: Wiley.

Irving, E. (1977). Drift of the major continental blocks since the Devonian. Nature, 270(5635), 304–309.

Irving, E., & Irving, G. A. (1982). Apparent polar wander paths Carboniferous through Cenozoic and the assembly of Gondwana. *Geophysical Surveys*, 5(2), 141–188.

Jarboe, N. A., Koppers, A. A., Tauxe, L., Minnett, R., & Constable, C. (2012). The online MagIC database: Data archiving, compilation, and visualization for the geomagnetic, paleomagnetic and rock magnetic communities. AGU Fall Meeting Abstracts (Vol. 2012, pp. GP31A-1063).

Johnson, C. L., Constable, C. G., Tauxe, L., Barendregt, R., Brown, L. L., Coe, R. S., et al. (2008). Recent investigations of the 0–5 Ma geomagnetic field recorded by lava flows. *Geochemistry, Geophysics, Geosystems*, 9, Q04032. https://doi.org/10.1029/2007GC001696

Jupp, P. E., & Kent, J. T. (1987). Fitting smooth paths to spherical data. Journal of the Royal Statistical Society: Series C (Applied Statistics), 36(1), 34–46.

#### Acknowledgments

BV and DJJvH acknowledge NWO Vici Grant 865.17.001. We thank Lisa Tauxe, Nick Swanson-Hysell and associate editor Daniel Pastor-Galán for their constructive feedback and helpful suggestions. We thank Cor Langereis and David Heslop for discussion. LCG has received funding from the European Union's Horizon 2020 research and innovation program under the Marie Skłodowska-Curie Grant Agreement No. 101025975.



Kent, D. V., & Irving, E. (2010). Influence of inclination error in sedimentary rocks on the Triassic and Jurassic apparent pole wander path for North America and implications for Cordilleran tectonics. *Journal of Geophysical Research*, 115, B10103. https://doi.org/10.1029/2009JB007205 Kent, D. V., & Muttoni, G. (2020). Pangea B and the late paleozoic ice age. *Palaeogeography, Palaeoclimatology, Palaeoecology*, 553(109753),

1-20. https://doi.org/10.1016/j.palaeo.2020.109753

King, R. F. (1955). The remanent magnetism of artificially deposited sediments. *Geophysical Supplements to the Monthly Notices of the Royal Astronomical Society*, 7(3), 115–134.

Kodama, K. P. (2012). Paleomagnetism of sedimentary rocks: Process and interpretation. John Wiley & Sons.

Koymans, M. R., Langereis, C. G., Pastor-Galán, D., & van Hinsbergen, D. J. J. (2016). Paleomagnetism.org: An online multi-platform open source environment for paleomagnetic data analysis. *Computers & Geosciences*, 93, 127–137.

Koymans, M. R., van Hinsbergen, D. J. J., Pastor-Galán, D., Vaes, B., & Langereis, C. G. (2020). Towards FAIR paleomagnetic data management through Paleomagnetism.org 2.0. Geochemistry, Geophysics, Geosystems, 21, e2019GC008838. https://doi.org/10.1029/2019GC00883

Kulakov, E. V., Torsvik, T. H., Doubrovine, P. V., Slagstad, T., Ganerød, M., Silkoset, P., & Werner, S. C. (2021). Jurassic fast polar shift rejected by a new high-quality paleomagnetic pole from southwest Greenland. *Gondwana Research*, 97, 240–262. https://doi.org/10.1016/j. gr.2021.05.021

Le Goff, M., Henry, B., & Daly, L. (1992). Practical method for drawing a VGP path. Physics of the Earth and Planetary Interiors, 70(3-4), 201-204.

McElhinny, M. W., Embleton, B. J. J., & Wellman, P. (1974). A synthesis of Australian Cenozoic palaeomagnetic results. *Geophysical Journal International*, 36(1), 141–151.

McElhinny, M. W., & McFadden, P. L. (1997). Palaeosecular variation over the past 5 Myr based on a new generalized database. *Geophysical Journal International*, 131(2), 240–252.

McElhinny, M. W., & McFadden, P. L. (2000). Paleomagnetism: Continents and oceans. Elsevier.

McEnroe, S. A. (1996). North America during the lower Cretaceous: New palaeomagnetic constraints from intrusions in new England. Geophysical Journal International, 126(2), 477–494.

McFadden, P. L., & McElhinny, M. W. (1995). Combining groups of paleomagnetic directions or poles. *Geophysical Research Letters*, 22(16), 2191–2194.

McFadden, P. L., Merrill, R. T., McElhinny, M. W., & Lee, S. (1991). Reversals of the Earth's magnetic field and temporal variations of the dynamo families. *Journal of Geophysical Research*, 96(B3), 3923–3933.

Meert, J. G., Pivarunas, A. F., Evans, D. A., Pisarevsky, S. A., Pesonen, L. J., Li, Z. X., et al. (2020). The magnificent seven: A proposal for modest revision of the quality index. *Tectonophysics*, 790, 228549. https://doi.org/10.1016/j.tecto.2020.228549

Morel, P., & Irving, E. (1981). Paleomagnetism and the evolution of Pangea. Journal of Geophysical Research, 86(B3), 1858–1872.

Musgrave, R. J. (1989). A weighted least-squares fit of the Australian apparent polar wander path for the last 100 Myr. *Geophysical Journal International*, 96(2), 231–243.

Muttoni, G., Dallanave, E., & Channell, J. E. T. (2013). The drift history of Adria and Africa from 280 Ma to Present, Jurassic true polar wander, and zonal climate control on Tethyan sedimentary facies. *Palaeogeography, Palaeoclimatology, Palaeoecology*, 386, 415–435.

Muttoni, G., & Kent, D. V. (2019). Jurassic monster polar shift confirmed by sequential paleopoles from Adria, promontory of Africa. Journal of Geophysical Research: Solid Earth, 124, 3288–3306. https://doi.org/10.1029/2018JB017199

Oliva-Urcia, B., Gil-Peña, I., Maestro, A., López-Martínez, J., Galindo-Zaldívar, J., Soto, R., et al. (2016). Paleomagnetism from Deception Island (South Shetlands archipelago, Antarctica), new insights into the interpretation of the volcanic evolution using a geomagnetic model. International Journal of Earth Sciences, 105(5), 1353–1370.

Phillips, J. D., & Forsyth, D. (1972). Plate tectonics, paleomagnetism, and the opening of the Atlantic. *The Geological Society of America Bulle*tin, 83(6), 1579–1600.

Riisager, J., Riisager, P., & Pedersen, A. K. (2003). The C27n-C26r geomagnetic polarity reversal recorded in the west Greenland flood basalt province: How complex is the transitional field? *Journal of Geophysical Research*, 108(B3), 2155. https://doi.org/10.1029/2002JB002124

Rowley, D. B. (2019). Comparing paleomagnetic study means with apparent wander paths: A case study and paleomagnetic test of the Greater India versus Greater Indian basin hypotheses. *Tectonics*, 38(2), 722–740.

Schettino, A., & Scotese, C. R. (2005). Apparent polar wander paths for the major continents (200 Ma to the present day): A palaeomagnetic reference frame for global plate tectonic reconstructions. *Geophysical Journal International*, 163(2), 727–759.

Smirnov, A. V., & Tarduno, J. A. (2010). Co-location of eruption sites of the Siberian Traps and North Atlantic igneous province: Implications for the nature of hotspots and mantle plumes. *Earth and Planetary Science Letters*, 297(3–4), 687–690.

Suttie, N., Biggin, A., & Holme, R. (2015). Robust estimators of palaeosecular variation. *Geophysical Journal International*, 200(2), 1046–1051.
Swanson-Hysell, N. L., Ramezani, J., Fairchild, L. M., & Rose, I. R. (2019). Failed rifting and fast drifting: Midcontinent Rift development, Laurentia's rapid motion and the driver of Grenvillian orogenesis. *Bulletin*, 131(5–6), 913–940.

Swanson-Hysell, N. L., Vaughan, A. A., Mustain, M. R., & Asp, K. E. (2014). Confirmation of progressive plate motion during the Midcontinent Rift's early magmatic stage from the Osler Volcanic Group, Ontario, Canada. *Geochemistry, Geophysics, Geosystems, 15*, 2039–2047. https:// doi.org/10.1002/2013GC00518

Tauxe, L., Banerjee, S. K., Butler, R. F., & van der Voo, R. (2010). Essentials of paleomagnetism. University of California Press.

Tauxe, L., Constable, C., Johnson, C. L., Koppers, A. A., Miller, W. R., & Staudigel, H. (2003). Paleomagnetism of the southwestern USA recorded by 0–5 Ma igneous rocks. *Geochemistry, Geophysics, Geosystems*, 4(4), 8802. doi: 10.1029/2002GC000343

Tauxe, L., & Kent, D. V. (2004). A simplified statistical model for the geomagnetic field and the detection of shallow bias in paleomagnetic inclinations: Was the ancient magnetic field dipolar? *Geophysical Monograph Series*, 145, 101–155.

Tauxe, L., Kylstra, N., & Constable, C. (1991). Bootstrap statistics for paleomagnetic data. Journal of Geophysical Research, 96(B7), 11723–11740.

Tauxe, L., Shaar, R., Jonestrask, L., Swanson-Hysell, N. L., Minnett, R., Koppers, A. A. P., et al. (2016). PmagPy: Software package for paleomagnetic data analysis and a bridge to the magnetics information Consortium (MagIC) database. *Geochemistry, Geophysics, Geosystems*, 17, 2450–2463. https://doi.org/10.1002/2016GC006307

Thompson, R., & Clark, R. M. (1981). Fitting polar wander paths. Physics of the Earth and Planetary Interiors, 27(1), 1-7.

Thompson, R., & Clark, R. M. (1982). A robust least-squares Gondwanan apparent polar wander path and the question of palaeomagnetic assessment of Gondwanan reconstruction. *Earth and Planetary Science Letters*, 57(1), 152–158.

Torsvik, T. H., Müller, R. D., Van der Voo, R., Steinberger, B., & Gaina, C. (2008). Global plate motion frames: Toward a unified model. *Reviews of Geophysics*, *46*, RG3004. https://doi.org/10.1029/2007RG000227

Torsvik, T. H., Smethurst, M. A., Meert, J. G., Van der Voo, R., McKerrow, W. S., Brasier, M. D., et al. (1996). Continental break-up and collision in the Neoproterozoic and Palaeozoic—A tale of Baltica and Laurentia. *Earth-Science Reviews*, 40(3–4), 229–258.



- Torsvik, T. H., Smethurst, M. A., Van der Voo, R., Trench, A., Abrahamsen, N., & Halvorsen, E. (1992). Baltica. A synopsis of Vendian-Permian palaeomagnetic data and their palaeotectonic implications. *Earth-Science Reviews*, *33*(2), 133–152.
- Torsvik, T. H., Van der Voo, R., Preeden, U., Mac Niocaill, C., Steinberger, B., Doubrovine, P. V., et al. (2012). Phanerozoic polar wander, palaeogeography and dynamics. *Earth-Science Reviews*, 114(3–4), 325–368.
- Vaes, B. (2022). Software for: "On pole position: Causes of dispersion of the paleomagnetic poles behind apparent polar wander paths. Zenodo. https://doi.org/10.5281/zenodo.6366399
- Vaes, B., Li, S., Langereis, C. G., & van Hinsbergen, D. J. (2021). Reliability of palaeomagnetic poles from sedimentary rocks. *Geophysical Journal International*, 225(2), 1281–1303.
- van Alstine, D. R., & de Boer, J. (1978). A new technique for constructing apparent polar wander paths and the revised Phanerozoic path for North America. *Geology*, 6(3), 137–139.
- Van der Voo, R. (1990). The reliability of paleomagnetic data. Tectonophysics, 184(1), 1-9.
- Van der Voo, R., & French, R. B. (1974). Apparent polar wandering for the Atlantic-bordering continents: Late Carboniferous to Eocene. *Earth-Science Reviews*, 10(2), 99–119.
- van Hinsbergen, D. J., de Groot, L. V., van Schaik, S. J., Spakman, W., Bijl, P. K., Sluijs, A., et al. (2015). A paleolatitude calculator for paleoclimate studies. *PLoS One*, 10(6), e0126946.
- van Hinsbergen, D. J., Spakman, W., Vissers, R. L., & van der Meer, D. G. (2017). Comment on "assessing discrepancies between previous plate Kinematic models of Mesozoic Iberia and their constraints" by Barnett-Moore et Al. *Tectonics*, 36(12), 3277–3285.
- Van Hinsbergen, D. J., Straathof, G. B., Kuiper, K. F., Cunningham, W. D., & Wijbrans, J. (2008). No vertical axis rotations during Neogene transpressional orogeny in the NE Gobi Altai: Coinciding Mongolian and Eurasian early Cretaceous apparent polar wander paths. *Geophysical Journal International*, 173(1), 105–126.
- Vandamme, D., Courtillot, V., Besse, J., & Montigny, R. (1991). Paleomagnetism and age determinations of the Deccan Traps (India): Results of a Nagpur-Bombay Traverse and review of earlier work. *Reviews of Geophysics*, 29(2), 159–190.
- Wu, L., Murphy, J. B., Quesada, C., Li, Z. X., Waldron, J. W., Williams, S., et al. (2021). The amalgamation of Pangea: Paleomagnetic and geological observations revisited. *Bulletin*, 133(3–4), 625–646.